Author’s response to reviews

Title: A qualitative analysis of men's involvement in maternal and child health as a policy intervention in rural Central Malawi

Authors:

Elizabeth Mkandawire (elimka23@gmail.com)

Sheryl Hendriks (sheryl.hendriks@up.ac.za)

Version: 3 Date: 05 Jan 2018

Author’s response to reviews:

Dear Dr Aronin,

Thank you for the good news on this paper. We have revised the attached version of the paper as requested. This includes removing the additional files, sending a clearer copy of the change of authorship form to Dr Medel (copied to you), removing the recommendations section and incorporating these into the text in the conclusions section, and editing the table to distinguish between the themes and sub-themes (shading has also been used to make this distinction clearer).

We thank you and the reviewers for the most helpful comments that have helped us to improve the quality of the paper.

Yours sincerely,

Sheryl and Elizabeth